Author’s response to reviews

Title: The multiple systemic artery to pulmonary artery fistulas resulting in severe irreversible pulmonary arterial hypertension in patient with previous history of pneumothorax.

Authors:

Wojciech Jacheć (wjachec@interia.pl)
Andrzej Tomasik (tomasik@poczta.onet.pl)
Marcin Kurzyna (marcin.kurzyna@ecz-otwock.pl)
Radosław Pietura (radoslawpietura@poczta.onet.pl)
Adam Torbicki (adam.torbicki@ecz-otwock.pl)
Jan Głowacki (j.glowacki@sccs.pl)
Ewa Nowalany-Kozielska (ewakozielska@wp.pl)
Celina Wojciechowska (wojciechowskac@wp.pl)

Version: 3 Date: 24 Feb 2019

Author’s response to reviews:

Dear Michele Mondoni

BMC Pulmonary Medicine

Thank you for your review. You can find our replies for the reviewers’ comments hereunder. We would like to inform you that they have been taken into account in the new version of the manuscript substantially.

Sincerely

On behalf of the co-authors

Wojciech Jacheć
Dear Andrea Lombi MD

Thank you very much

Yours faithfully

on behalf of co-authors

Wojciech Jacheć

Dear Rocco Francesco Rinaldo, MD

Thank you very much for the effort put in reviewing the manuscript. As suggested, we made the following changes:

The results of cardiac catheterization have been supplemented (Table 2 – control RHC).

The phrase "normal PCWP" was removed.

The final conclusion contained in the chapter "Take home message" has been changed.

Yours faithfully

on behalf of co-authors

Wojciech Jacheć